Author's response to reviews

Title: Type II Collagen and Glycosaminoglycan expression induction in primary human chondrocyte by TGF-beta1

Authors:

Hyun Joo Yoon (hyoon@tissuegene.com)
Suk Bum Kim (bum3340@hanmail.net)
Dhara Somaiya (dsomaiya@tissuegene.com)
Moon Jong Noh (mnoh@tissuegene.com)
Kyoung-Baek Choi (klsho@kolon.com)
Chae-Lyul Lim (cllim@kolon.com)
Hyeon-Youl Lee (yalli74@kolon.com)
Yeon-Ju Lee (csgr@kolon.com)
Youngsuk Yi (yiyounsguk@yahoo.com)
Kwan Hee Lee (ortholee@tissuegene.com)

Version: 2
Date: 29 December 2014

Author's response to reviews:

Dear Editor of BMC Musculoskeletal Disorders,

Thank you for your pre-review comment and advice.

As requested, the manuscript has changed.

First, page numbers and line numbers were added on the main manuscript.

Second, the agreement sentence from cell donor was included in the method section.

Please let me know if there is any additional change required.

Best regards,

Hyun Joo Yoon